Author's response to reviews

Title: Pedigree analysis and epidemiological features of idiopathic congenital talipes equinovarus in the United Kingdom: A case-control study

Authors:

Amanda H Cardy (a.h.cardy@abdn.ac.uk)
Simon Barker (s.barker@nhs.net)
David J Chesney (dchesney@doctors.org.uk)
Linda Sharp (linda.sharp@ncri.ie)
Nicola Maffulli (n.maffulli@orth.keele.ac.uk)
Zosia Miedzybrodzka (zosia@abdn.ac.uk)

Version: 4 Date: 28 June 2007

Author's response to reviews:

Changes as requested in your email of 26th June have been made:
Change to title, MS and on submission system
'Author affiliation' removed
Author's contributions changed to Authors' contributions
Titles of tables placed above the tables
Blank page removed